Author's response to reviews

Title: Prolapse of fallopian tube through abdominal wound after caesarean section mimicking scar endometriosis- an unusual case report.

Authors:

Lajya D Goyal (lajja.goyal@rediffmail.com)
Shalini Maheshwari (drshalini2000@gmail.com)
Sharanjit Kaur (jitsharan28@gmail.com)
Harpreet Kaur (hksobgyn35@yahoo.co.in)

Version: 2 Date: 17 March 2015

Author's response to reviews: see over
To
The Editor

Sub: Submission of Manuscript for publication

Dear Sir,

We intend to publish a case report entitled “Prolapse of fallopian tube through abdominal wound after caesarean section mimicking scar endometriosis - an unusual case report” in your esteemed journal.

We are reporting a case of prolapsed fimbrial end of the fallopian tube through abdominal scar after caesarean section which mimicked scar endometriosis which is very unusual. This is the first case report best to our knowledge on fallopian tube prolapsed through abdominal wound after caesarean section.

This report will add one more differential diagnosis where the suspiciousness arises for scar endometriosis.

This is a original case report that contribute to the field of obstetrics and gynaecology and women health.

Prior Publications: Nil
Support: Nil
Conflict of Interest: Nil

We hereby transfer, assign, or otherwise convey all copyright ownership, including any and all rights incidental thereto, exclusively to the journal, in the event that such work is published by the journal.

As per your suggestion, I have made all the required changes and revised the manuscript as per journals format.

Kindly let me know if any more modifications are required.

Yours’ sincerely,
Sharanjit Kaur

Cooresponding Author:
Sharanjit Kaur, Lecturer, University College of Nursing, Baba Farid University of Health sciences, Faridkot, Pin Code 151203
jitsharan28@gmail.com